Author’s response to reviews

Title: Influence of Indoor Work Environments on Health, Safety, and Human Rights among Migrant Sex Workers at the Guatemala-Mexico Border: A Call for Occupational Health and Safety Interventions

Authors:
Shira Goldenberg (gshi-sg@cfenet.ubc.ca)
Teresita Rocha Jimenez (trochaji@ucsd.edu)
Kimberly Brouwer (kbrouwer@ucsd.edu)
Sonia Morales Miranda (sonia.moralesmiranda30@gmail.com)
Jay Silverman (jgsilverman@ucsd.edu)

Version: 2 Date: 17 Jan 2018

Author’s response to reviews:

RESPONSE TO REVIEWERS

Reviewer Reports:

Reviewer 1: The authors have done a great job in strengthening the paper based on feedback. I believe that the paper is ready for publication.

We appreciative this positive feedback from Reviewer 1. Thank you again for the constructive feedback!

Editor Comments:

(1) Please provide the email addresses of all authors within the title page of your manuscript.

We apologize for the oversight. This information has been added to our title page.

(2) Please revise your 'Consent to publish' section. If your manuscript contains any individual person’s data in any form (including individual details, images or videos), consent for publication must be obtained from that person, or in the case of children, their parent or legal guardian. All presentations of case reports must have consent for publication. If your manuscript does not contain data from any individual person, please state “Not applicable” in this section.
Thank you, we have clarified that this is ‘not applicable’ under the consent to publish section.

(3) Please state at the end of your 'Authors Contributions' section that 'All authors read and approved the final manuscript.'

Thank you, this has been added to the ‘Author Contributions’ section.